Author's response to reviews

Title: Complementary and conventional medicine: a concept map

Authors:

Kendall Kroesen (kkroesen@qwest.net)
Dr Carol M. Baldwin (carolb@resp-sci.arizona.edu)
William M. Trochim (wmt1@cornell.edu)
Dr Iris R. Bell (ibell@u.arizona.edu)

Version: 5 Date: 30 Dec 2003

Changed capitalizations as requested and transferred figures into jpg files.